Author's response to reviews

Title: Melatonin prevents morphine-induced hyperalgesia and tolerance in rats: role of protein kinase C and N-methyl-D-aspartate receptors

Authors:

Li Song (huaxisongli@hotmail.com)
Chaoran Wu (chaoranwu@aliyun.com)
Yunxia Zuo (zuoyunxiahxa@qq.com)

Version: 5
Date: 4 January 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for the email and the attached editor’s comments. We greatly appreciate that you give us this opportunity to revise the manuscript.

We entirely agree with the comments and have revised the manuscript accordingly.

A point-by-point response to editor’s comments is attached in separate pages. Changes have been made in the revised manuscript. We greatly appreciate your consideration. Many thanks.

Best regards

Yunxia Zuo, M.D., Ph.D.
Professor of Anesthesia Research
West China Hospital
Sichuan University
A point-by-point response to the editor’s comments
We thank for the editor’s comments. The following is a point-by-point response to each comment and changes have been made in the revised manuscript with red characters. The editor’s comments are in an italicized font and our responses in a bold font.

Editor’s comment:

Although the authors have done a good job of answering the reviewers’ comments, the manuscript still requires the following mandatory revisions:
1. Change each occurrence of the phrase "morphine tolerance to the analgesic effect" to "tolerance to morphine’s analgesic effect"
   Response: Yes, we have revised the phrase "morphine tolerance to the analgesic effect" to "tolerance to morphine’s analgesic effect" in all the manuscript with red characters.

2. Change the subtitle: "Melatonin rescued morphine tolerance" to "Morphine reduced morphine tolerance"
   Response: Yes, we have revised the subtitle: "Melatonin rescued morphine tolerance" to "Melatonin rescued morphine tolerance" P10L10 with red characters.